Author's response to reviews

Title: Concomitant Ipsilateral Subcapital and Intertrochanteric Fractures of the Femur: a case report.

Authors:

George Loupasis (geoloup@panafonet.gr)
Panagiotis G Ntagiopoulos (ntagiopoulos@hotmail.com)
Antonios Asimakopulos (ntagiopoulos@hotmail.com)

Version: 6 Date: 9 July 2010

Author's response to reviews: see over
Dear Editor:

Per Journal instructions we are submitting a revised manuscript for the following paper: MS 6437362543826525, ‘Concomitant Ipsilateral Subcapital and Intertrochanteric Fractures of the Femur: a case report.’ Authors: George Loupasis, Panagiotis G Ntagiopoulos and Antonios Asimakopoulos, and we hope to address the reviewers’ and the Editorial Office’s comments.

Reviewer #1:

The patient was riding his motorcycle at an approximate speed of 70 mph. He collided head on with a car causing him to be thrown from the motorcycle landing forcefully on his left outstretched leg.

The sentence has now been changed as suggested.

We hope that we have addressed the reviewers’ comments and that we have answered to your questions.

We are very glad that our work is being considered for publication by your esteemed journal and we are looking forward to hearing from you.

Regards,

PG Ntagiopoulos, MD